Author’s response to reviews

Title: Different responses of weather factors on hand, foot, and mouth disease in three different climate areas of Gansu, China

Authors:

Faxiang Gou (goufx2015@163.com)
Xinfeng Liu (liuxf606@126.com)
Jian He (1014770674@qq.com)
Dongpeng Liu (tonpern@sina.com)
Yao Cheng (398510190@qq.com)
Haixia Liu (262716409@qq.com)
Xiaoting Yang (gscdcsci@sina.com)
Kunfu Wei (326274842@qq.com)
Yunhe Zheng (232779356@qq.com)
Xiaojun Jiang (42555005@qq.com)
Lei Meng (ccdsusci@163.com)
Wenbiao Hu (gscdc901@163.com)

Version: 2 Date: 12 Nov 2017

Author’s response to reviews:

Reviewer 1:

Thanks for your kindly reminder. We have also checked the whole manuscript and corrected all typos that we could find.